Author's response to reviews

Title: Relationship between tooth loss and mortality in 80-year-old Japanese community-dwelling subjects

Authors:

Toshihiro Ansai (ansai@kyu-dent.ac.jp)
Yutaka Takata (yutaka@kyu-dent.ac.jp)
Inho Soh (i-soh@kyu-dent.ac.jp)
Shuji Awano (awa-shu@kyu-dent.ac.jp)
Akihiro Yoshida (akihiro@kyu-dent.ac.jp)
Kazuo Sonoki (sonoki@kyu-dent.ac.jp)
Tomoko Hamasaki (tomotalk@kwuc.ac.jp)
Takehiro Torisu (r07torisu@fa.kyu-dent.ac.jp)
Akira Sogame (sogame-a1171@pref.fukuoka.lg.jp)
Naoko Shimada (naoko_shimada01@city.kitakyushu.lg.jp)
Tadamichi Takehara (takehara@kyu-dent.ac.jp)

Version: 6 Date: 30 June 2010

Author's response to reviews: see over
June 30, 2010

The BMC Series Editorial Production Team

Dear Sir,

Thank you very much for your e-mail concerning formatting changes of our manuscript (MS: 8284165272983197). Changes in our manuscript are shown following.

Yours sincerely,

Toshihiro Ansai, D.D.S., Ph. D.
Division of Community Oral Health Science
Dept. of Health Promotion
Kyushu Dental College
Kokurakita-ku,
Kitakyushu 803-8580
JAPAN
Fax: +81-93-591-7736
E-mail: ansai@kyu-dent.ac.jp

Major revisions:
1. Heading of Conclusions have been added.

Minor revisions:
1. Highlighting has been removed from the text.
2. Percentages of Table 1 (Daily alcohol consumption) has been recalculated and revised.
3. Formatting of Table 2 has been revised.